Author's response to reviews

Title: An Empirical Study to Determine Factors That Motivate and Limit The Implementation Of ICT In Healthcare Environments

Authors:

Raj Gururajan (<u>gururaja@usq.edu.au</u>) Abdul Hafeez-Baig (<u>abdulhb@usq.edu.au</u>)

Version: 4 Date: 17 September 2013

Author's response to reviews:

we have updated details on ethics approval. we have listed the institutions that provided ethics approval for this study. The practice in India and in Pakistan is to obtain through the medical directors, and this was followed with a formal request, i the form of a letter. In Australia, the hospital provided approval as this was part of a major study conducted with the help of this hospital.